Author's response to reviews

Title: How satisfied are students with peer teaching in clinical communication skills?

Authors:

   Jonathan KA Mills (jon_mills89@hotmail.co.uk)
   William J Dalleywater (wdalleywater@googlemail.com)
   Victoria Tischler (victoria.tischler@nottingham.ac.uk)

Version: 2 Date: 30 September 2013

Author's response to reviews:

A statement has been added to clarify that ethics approval was not required as the study was an evaluation of teaching not involving patients. Students did not complete the questionnaire if they did not wish to participate in the study. All students received teaching as normal whether or not they completed the questionnaire.